Author's response to reviews

Title: Treatment of scoliosis in patients affected with Prader-Willi syndrome using various different techniques

Authors:

Tiziana Greggi (tiziana.greggi@ior.it)
Kostas Martikos (kostasmartikos@hotmail.com)
Francesco Lolli (francesco.lolli@fastwebnet.it)
Georgios Bakaloudis (bakaloudisgeorgios@hotmail.com)
Mario Di Silvestre (mario.disilvestre@ior.it)
Alfredo Cioni (alfredo.cioni@ior.it)
Giovanni Barbanti Bròdano (giovanni.barbantibrodano@ior.it)
Stefano Giacomini (stefano.giacomini@ior.it)

Version: 5 Date: 1 June 2010

Author's response to reviews:

Following Dr. Grivas' suggestion, we have added another two comments to the ms sent on May 10 and they concern
- the status of arthodesis or fusion
- the type of bone graft used in the patients.
Now, marked in red are all the modifications made to the ms.